Institute for Monetary and Financial Stability

HASAN DOLUCA ROMAN INDERST UFUK OTAG

Banking Competition and Risk-Taking When Borrowers Care about Financial Prudence

> Institute for Monetary and Financial Stability JOHANN WOLFGANG GOETHE-UNIVERSITÄT FRANKFURT AM MAIN

> > WORKING PAPER SERIES NO. 28 (2009)

PROF. DR. HELMUT SIEKMANN (HRSG.)

INSTITUTE FOR MONETARY AND FINANCIAL STABILITY PROFESSUR FÜR GELD-, WÄHRUNGS- UND NOTENBANKRECHT JOHANN WOLFGANG GOETHE-UNIVERSITÄT GRÜNEBURGPLATZ 1 60629 FRANKFURT AM MAIN

 TELEFON:
 (069) 798 – 34014

 TELEFAX:
 (069) 798 – 33913

 E-MAIL:
 GELD-UND-WAEHRUNG@IMFS-FRANKFURT.DE

HASAN DOLUCA ROMAN INDERST UFUK OTAG

Banking Competition and Risk-Taking When Borrowers Care about Financial Prudence

> Institute for Monetary and Financial Stability JOHANN WOLFGANG GOETHE-UNIVERSITÄT FRANKFURT AM MAIN

> > WORKING PAPER SERIES NO. 28 (2009)

Bank Competition and Risk-Taking When Borrowers Care about Financial Prudence

Hasan Doluca^{*} Roman Inderst[†] Ufuk $Otag^{\ddagger}$

First complete version August 2008; This version Mai 2009

Abstract

Corporate borrowers care about the overall riskiness of a bank's operations as their continued access to credit may rely on the bank's ability to roll over loans or to expand existing credit facilities. As we show, a key implication of this observation is that increasing competition among banks should have an *asymmetric* impact on banks' incentives to take on risk: Banks that are already riskier will take on yet more risk, while their safer rivals will become even more prudent. Our results offer new guidance for bank supervision in an increasingly competitive environment and may help to explain existing, ambiguous findings on the relationship between competition and risk-taking in banking. Furthermore, our results stress the beneficial role that competition can have for financial stability as it turns a bank's "prudence" into an important competitive advantage.

^{*}University of Frankfurt, IMFS. E-mail: doluca@finance.uni-frankfurt.de.

[†]University of Frankfurt, IMFS, and Imperial College London. E-mail: inderst@finance.unifrankfurt.de.

[‡]University of Frankfurt, IMFS. E-mail: otag@finance.uni-frankfurt.de.

1 Introduction

Over the last decades, in many countries the banking industry has become increasingly deregulated. This and the spread of financial innovations has widened the options that are available to commercial borrowers, leading to more intense competition among banks as well as between banks and alternative sources of finance.¹ Borrowers are, however, well advised to look further ahead when tapping into cheaper sources of (short-term) finance: Will they later be able to refinance a maturing loan and at what conditions? Will the existing lender roll over the loan, extend a credit facility, or provide additional finance at short notice? In particular for businesses that are less mature, smaller, or more opaque to investors, it could become quite costly (or even impossible) to replace an existing lending relationship at short notice, given that in this case the underlying relationship capital would be lost.²

Empirically, this is documented, for instance, in Slovin, Sushka, and Polonchek (1993), who found that borrowers from Continental Illinois suffered an average 4.2% loss in their stock market value after the bank failed. Likewise, Djankov, Jindra, and Klapper (2005) show that bank closures in Indonesia, Korea, and Thailand decreased borrowers' stock prices by 3.9%, while Yamori and Murakami (1999) document a 6.6% decrease for those borrowers who named the failed Japanese bank, Hokkaido Takusyoku, as their main lender.

While outright bankruptcy is an extreme event, borrowers may also be adversely affected if liquidity problems force their main lender to call back loans or to refuse the expansion of existing credit facilities. To the extent that an existing relationship involves some degree of "lock-in", e.g., due to an informational advantage of an existing lender, the borrower may not receive adequate funding elsewhere.

Based on these observations, this paper starts from the presumption that the riskiness of a bank's existing operations, as well as its leverage, represent a key *quality* attribute in the eyes of potential borrowers. Banks that are perceived as being less aggressive in undertaking (on- and off balance sheet) risk would then be regarded by borrowers as a superior choice. As we show, this can play an important role in making banks more conservative. In fact, unless they are levered up sufficiently, e.g., through their deposit-

¹On the latter see, in particular, Boot and Thakor (2000). While this process has clearly lost momentum in the present (as of spring 2009) financial crisis, it remains to be seen whether or to what extent it will be reverted after all.

²This is a key notion in the large literature on relationship lending (cf. Boot (2000)).

taking activities, they may even choose to forego positive-NPV "gambles". The focus of this paper is, however, on the interaction of risk-taking and changes in competition.

We find that competition (e.g., through deregulation that limits the scope for horizontal differentiation) has an *asymmetric* impact on banks' risk-taking incentives. As competition becomes more intense, it is likely that some banks become more prudent, while their rivals undertake riskier strategies. In particular, if banks already differ in the riskiness of their existing operations, more competition induces less (additional) risk taking from an already more prudent bank, while it has the opposite effect on its already riskier rivals.

This finding may provide some guidance to bank supervision in an increasingly competitive environment. Our results would support neither the view that supervision must be *uniformly* stepped up as competition increases nor the view that competition *uniformly* reduces the need for supervision. Instead, as competition for loans increases, supervisory activity should become more focused on those banks that are already riskier, while oversight could be relaxed for institutions that are already more prudent (that is, provided that the authority's objective is indeed to identify and potentially limit (additional) risk taking). Furthermore, we also show that unless they are levered up sufficiently, banks may even be too conservative if their perceived future liquidity represents a main competitive variable. To obtain this result, the strategic interaction of banks in the market for borrowers is important.³

As competition increases, we thus identify a tendency towards more "vertical differentiation", given that in our model banks' commercial borrowers care about how much risk banks take on through other operations.⁴ The asymmetry in risk-taking behavior that this paper identifies is also supported by recent evidence in Gropp, Hakenes, and Schnabel (2006). Though there the reason for why some banks become "less risky" is exogenous, as it depends on government guarantees, their finding that these guarantees "strongly increase the risk-taking of the competitor banks" would be predicted by our model.

³Put differently, our result does not simply follow from the observation that banks may remain prudent so as to take advantage of future profit-making opportunities.

⁴This tendency may counteract the risk of "herding" that has been identified in other papers. For instance, such herding may follow from the expectation that the regulator will more likely bail out individual banks if more of them find themselves in a crisis (cf. Acharya and Yorulmazer (2007) for such a "too-many-to-fail" argument).

Relation to the Literature. Our finding that competition may have an asymmetric impact on banks' willingness to take on more risk mirrors the ambiguous findings in the empirical literature. In a recent study, Beck, Kunt, and Levine (2003) find that different measures of competition, e.g., based on concentration or the degree of deregulation, predict different relationships between competition and the stability of the banking industry. Boyd, De Nicolo, and Jalal (2006) have shown that a more concentrated banking sector may be associated with less stability (cf. also De Nicolo and Loukoivanova (2007)). This contrasts with earlier work as well as, more recently, with the findings of Jiménez et al. (2007) for a large sample of Spanish banks (cf. also the references therein). Our results suggest that competition can indeed have mixed implications, though we provide predictions for which banks should be expected to engage in more risk taking and which banks should not.

In terms of theory, most of the extant literature would assert a *positive* correlation between competition and banks' incentives to take on (more) risk. Following Keeley (1990), one suggested channel works through a reduction in banks' charter value.⁵ Perotti and Suarez (2002), instead, find by comparing a monopoly with a banking duopoly that in the latter case banks are more prudent as either bank hopes to enjoy monopoly profits by being "the last bank standing", once the other bank has failed.

It should also be noted that our paper focuses on banks' *incentives* to take on risk. Clearly, holding all else constant, a reduction in banks' profitability may also mechanically imply a higher risk of becoming insolvent. Another channel through which competition affects banks' riskiness *without* affecting their own risk-taking incentives was identified in Boyd and Nicolo (2005), where a lower interest rate induces less risk taking by *borrowers*, which in turn makes loans and thus banks' balance sheets less risky (cf. also more recently Martinez-Miera and Repullo (2008) for an extension and qualification of their results). On the other hand, Caminal and Matutes (1997) and Cordella and Yeyati (2002) argue that under more intense competition banks would monitor less, thereby increasing the

⁵This has been confirmed and extended by a number of other papers using dynamic models, e.g., Suarez (1994), Hellmann et al. (2000), or Repullo (2004), to name only a few.

likelihood of bad loans.⁶

Our model departs from the established literature in a number of ways. First, the mechanism that links competition to risk-taking in our model is novel. In fact, to our knowledge the relevance of banks' own riskiness for *borrowers* has been largely ignored in the (theoretical) literature.⁷ Second, our implications are different from those in the extant literature, which would typically predict a symmetric response of all banks to more competition.⁸ Finally, while some of the arguments in the extant literature apply indiscriminately of whether there is more competition on the asset *or* liability side (say, for borrowers or depositors), e.g., given that they hinge simply on banks' lower overall profitability, our channel, which predicts an asymmetric change in risk-taking incentives, works exclusively through more competition in the markets for loans.

Organization of the Paper. The rest of this paper is organized as follows. Section 2 introduces the model. In Section 3 we solve the baseline model and derive first results, albeit restricting attention to *ex-ante* symmetric banks. Section 5 analyzes the interaction of risk taking and competition, while Section 6 explores the role of (higher) leverage. Section 7 introduces *ex-ante* asymmetry to derive more implications. Section 8 compares the market outcome to a benchmark of efficiency. We conclude in Section 9.

2 The Model

We consider competition between two banks, indexed by i = 1, 2. For the purpose of this paper we take each bank's capital structure as given. For instance, as frequently done in the banking literature, we may suppose that it is determined by a bank's access

⁶In a similar vein, competition reduces the incentives to screen borrowers in Chan et al. (1986) or leads to more risk-taking by eroding informational rents originating from relationship banking in Besanko and Thakor (1993). See further Matutes and Vives (1996, 2000) for models of imperfect competition generating a positive relationship between competition and risk (as well as Carletti and Hartmann (2003) for a recent survey).

⁷For an exception see Detragiache, Garella, and Guiso (2000), which seeks to explain the number of banking relationships that a firm entertains.

⁸The asymmetry is, however, shared with recent work by Boot and Marinč (2006). There, banks differ in their ability to monitor loans but can undertake investments to raise their overall level of monitoring. As competition becomes more intense, those banks that have a lower intrinsic ability to monitor expect to obtain a lower market share, which given the lump-sum nature of any investment into their monitoring capacity reduces their investment incentives. For banks with a higher intrinsic ability to monitor the opposite prediction holds.

to (cheap) deposit finance. We suppose that originally each bank has capital k > 0, of which some fraction was financed by debt and deposits. What matters for a bank's risktaking incentives is the total amount that the bank will have to repay to debtholders before equityholders are paid off. We suppose that the bank is run in the interest of equityholders. We let D_i denote a bank's total repayment obligations.

We will later allow banks to differ in the riskiness of their existing business. For the time being, however, we still abstract from this and take also banks' capital structure as exogenously given.

The game then unfolds from time t = 1 onwards, where banks can choose a potentially risky activity. This could represent, for instance, proprietary trading, additional loan commitments (e.g., as back-up financing), guarantees, or other contingent obligations. For simplicity, we stipulate that initially this activity, to which we refer to as activity A, does not require capital. Both banks have access to it. The outcome of this activity is known to the bank at the end of period t = 1 and leads to payments in t = 2. With probability pthe bank will realize from activity A the payoff z' > 0, while with the residual probability 1 - p the payoff z'' < 0 materializes. In other words, with probability 1 - p the bank will have a net outflow of z''.

It is convenient to denote F := |z'' - z'| and f =: z'. This allows us to analyze activity A in terms of some "fee" f that the bank receives for the risk of having to pay out F with probability 1 - p.⁹ As we abstract from discounting and as all parties are risk neutral, the net present value of activity A is thus

$$\eta_A := f - (1 - p)F.$$

We stipulate that whether a bank chooses the risky activity A or not is observable.¹⁰ This will prove to be an important information for corporate borrowers, for which banks subsequently compete. For notational simplicity it is convenient to stipulate that it is still in period t = 1 that banks compete in the loan market.

Competition. Crucially, our set-up for the loan market must allow to deal with various degrees of competition in a tractable way. This is accomplished by using a standard model

⁹For the purpose of our analysis it will be inconsequential whether and to what extent the outcomes of the two banks' risky activities are correlated.

¹⁰Though our results extend to the case where this is only observed with some noise, the resulting signaling game would heavily complicate the analysis.

of horizontal differentiation, namely that of Hotelling competition. Here, differentiation could capture the extent to which a bank's more "local" customers are both able and willing to choose a more distant competitor. The associated costs (both for lenders and borrowers) could be influenced both by regulation and by changes in the lending technology.

To be precise, we stipulate that there is a single potential borrower whose preferences regarding the choice between the two banks are *ex-ante* unknown.¹¹ His preferences are captured by a variable x that is uniformly distributed over $x \in [0,1]$ such that when borrowing from bank i = 1, the borrower will incur a disutility (measured in units of money) of τx , while when borrowing from bank i = 2 the respective disutility will be $(1-x)\tau$. One standard (Hotelling) interpretation is in terms of "shoe leather" costs, with x representing a measure of the distance between the bank's and the borrower's premises.¹²

The loan contract prescribes that the borrower obtains capital k, which is fully invested into a long-term project. If the project is continued until its end, which is in t = 3, then it pays off y with probability q and zero otherwise. The expected payoff qy is supposed to exceed k. If liquidated prematurely in period t = 2, however, the project's payoff is only $\beta y < k$.

Loan Contract. The loan contract offered by bank *i* prescribes a total repayment of R_i comprising the principal *k* and interest kr_i if the project is continued until t = 3. If the project is terminated prematurely in t = 2, all of its liquidation value $\beta y < k$ is seized by the bank. Importantly, note that the loan contract is only short term, allowing the bank to recall the loan in t = 1.¹³ The specifications of the loan contract deserve some additional comments.

The fact that the contract is only short term is realistic. It is also easily endogenized by appealing to agency problems, either of moral hazard or adverse selection, between the borrower and the bank. For instance, we could imagine that the borrower can be of two types $\xi = l, h$. The type is only privately known to the borrower originally, but revealed

¹¹We could imagine that among a number of borrowers who are located at different distances to each bank only one may have financing needs and that his identity is *ex-ante* not known.

 $^{^{12}}$ See, for instance, Degryse and Ongena (2005) for a recent empirical application or Degryse et al. (2006) for a wider interpretation of this approach. Our results would also follow from a model using a Salop circle, in which case increased competition could also arise from an increase in the number of competitors that are distributed uniformly over the circle.

¹³For instance, this could be a short-term revolving loan facility or a revocable overdraft facility that is drawn down fully in t = 1.

to the bank in t = 2. Only a project of type $\xi = h$ has a chance of realizing a payoff yin t = 3, namely with probability q_h , while the respective probability of a type-l project is $q_l = 0$. In addition, borrowers realize (arbitrarily small but strictly positive) private benefits in case the project is financed and continued until t = 3. Facing such a problem of adverse selection, the bank would want to preserve the right to call back the loan in t = 1 and, thereby, obtain at least the value of the seized assets βy . Otherwise, it would risk attracting also type-l borrowers.¹⁴

Banks' Capitalization. In our model, the bank may call back the loan early so as to prevent its own insolvency. In fact, this will be the case if the bank previously undertook the risky activity and if the underlying gamble resulted in an obligation to pay out F in t = 2. (Note that we assume here also that $\beta y \ge F$.) As calling back the loan is clearly inefficient given that $qy > k > \beta y$, one may ask why the bank does not either provide for more capital up-front, namely at least (F - f) + k, or raise capital in t = 2 so as to absorb the loss of F. Though these clearly represent feasible options, realistically both would have drawbacks on their own.

First, if the firm raised more than capital k up-front, then this could give rise to ("free cash-flow") agency problems between the bank and its investors.¹⁵ Formally, we could imagine that the bank's managers consume a fraction $\gamma > 0$ of all funds that have not been used up until t = 2. As γ becomes sufficiently large, it would not be optimal to raise more than k up-front. Raising additional capital F in t = 2 may also prove too expensive as new investors may be at an informational disadvantage ("dilution problem"). Formally, we may suppose that at the end of t = 1 the bank, possibly together with the borrower, learns about some features of the investment project, e.g., as captured by some type $\xi = l, h$. When the bank raises new capital on the back of its existing assets, a bank with a (relatively) bad loan may also want to raise new capital even without having immediate liquidity needs (cf. Stein (1998), Winton (2003)).¹⁶

¹⁴Strictly speaking, we would also need that the pool of type-l borrowers (i.e. of so-called fraudulent applicants or "fly-by-night operators") is sufficiently large.

¹⁵This or another reason for why bank capital is costly needs to be invoked also in other models where banks are capital constrained at some *interim* stage, i.e., where banks did not keep sufficiently large buffer capital initially.

¹⁶A similar argument of adverse selection can be used to rule out the possibility that when having to repay his loan early, the borrower can successfully turn to another investor.

Comment: A Model of Banks. In our main analysis we set banks' leverage sufficiently low such that this by itself does not generate additional risk-taking incentives, allowing us to focus on the novel contribution of our model. We show subsequently that our equilibrium characterization survives if leverage adds additional risk-taking incentives (albeit, as is well known, expressions then quickly become unwieldy).

Still, we believe that our model and analysis is one that deals genuinely with risktaking in banking rather than, say, risk-taking in any non-financial corporation. This is the case as the "knock-on" effect of the bank's additional risky activity works through its commercial lending operation, where potential borrowers care about the overall soundness of the lender's financial position.¹⁷

3 Equilibrium

As noted previously, we first derive results for the benchmark case where the level of outstanding debt, including deposits, does not itself affect banks' risk-taking incentives. Hence, for the moment we can thus presume that the bank operates to maximize total firm value. Abstracting thereby from (more standard) risk-taking incentives that follow from leverage, this allows to focus on the novel mechanism that is at work in our model.

As also noted above, in our baseline case banks have no other ongoing activities in place. Consequently, the first decision is made at the beginning of t = 1, where banks can choose whether to undertake the risky activity A. Subsequently, they compete for the borrower by making offers R_i . In what follows, we proceed backwards by first solving for the equilibrium in the loan market. Subsequently, we determine the equilibrium at the initial stage, where the risky activity A can be undertaken.

3.1 Competition for Loans

Banks compete for the borrower at the end of period t = 1. For the moment, it is convenient to denote more generally by p_i the commonly known probability with which

¹⁷Having said this, the literature has also drawn attention to cases where other "stakeholders" such as customers or employees should be concerned about a firm's balance sheet, at least if there is some (relationship-specific) lock-in. Arguably, in a relationship between a lender and a commercial borrower we can rightly assume that the latter is sufficiently sophisticated to learn from analysts whether the former is relatively more or less in good shape compared to its peers. In addition, the risk imposed on the borrower in case of a subsequent solvency problem of its main bank could be large (as, for instance, evidenced by the findings that were discussed in the Introduction).

bank *i* will be able to roll over the loan in t = 1. (In equilibrium, we will thus have that $p_i \in \{p, 1\}$.)

A borrower who is located at $x \in [0, 1]$ and takes out a loan with bank i = 1 realizes the expected payoff

$$p_1 q(y - R_1) - x\tau. \tag{1}$$

Note that (1) takes into account that the loan will be recalled with probability $1 - p_1$. If the loan is, instead, continued, then the borrower realizes with probability q the project's payoff y minus the contractually stipulated repayment R_1 . Finally, the last term $x\tau > 0$ in (1) captures the degree of banks' horizontal differentiation.

In analogy to (1), the borrower's expected payoff when turning to bank i = 2 equals

$$p_2q(y-R_2) - (1-x)\tau.$$
 (2)

Comparing (1) with (2), the borrower is just indifferent between the two offers if x is equal to some critical value \hat{x} given by¹⁸

$$\widehat{x} = \frac{1}{2} + \frac{p_1 q(y - R_1) - p_2 q(y - R_2)}{2\tau}.$$
(3)

Note that \hat{x} is also the *ex-ante* probability with which, for given (p_i, R_i) , bank i = 1 will attract the borrower, while the respective *ex-ante* probability for bank i = 2 equals $1 - \hat{x}$.

Using \hat{x} , the expected profit of bank i = 1 from the loan business is thus given by

$$\hat{x} \left[p_1 q R_1 + (1 - p_1) \beta y - k \right].$$
(4)

As we so far assume that the bank's own leverage does not affect decision making, the optimal choice of R_1 maximizes (4). Likewise, the optimal choice of R_2 for the rival bank i = 2 maximizes

$$(1 - \hat{x}) \left[p_2 q R_2 + (1 - p_2) \beta y - k \right].$$
(5)

By assuming that

$$(1-p)y(q-\beta) < 3\tau \tag{6}$$

holds, we can ensure for $p_i \in \{p, 1\}$ that in equilibrium both banks attract the borrower with positive probability as $0 < \hat{x} < 1$. From the respective first-order conditions for (4) and from (5) we have the following result.

¹⁸It should be noted that (3) only applies if it satisfies $0 \leq \hat{x} \leq 1$, which in turn is the case if $|p_1q(y-R_1)-p_2q(y-R_2)| \leq \tau$ holds. Otherwise, we clearly have that either $\hat{x} = 0$ or $\hat{x} = 1$. Below we will invoke a condition that ensures that the solution is indeed interior.

Proposition 1 For a given risk profile, as captured by the respective values of p_i , banks offer loans that stipulate a required repayment of

$$R_{i} = \frac{1}{qp_{i}} \left[k + \tau - y\beta(1 - p_{i}) + \frac{y(p_{i} - p_{j})(q - \beta)}{3} \right].$$
 (7)

Expression (7) is intuitive. Note first that if neither bank engages in the risky activity A such that $p_1 = p_2 = 1$, then this simplifies to $R_1 = R_2 = R$ with $qR = k + \tau$. In other words, the *expected* total repayment, qR, is equal to the principal k plus a margin that is equal to the measure of horizontal differentiation, τ . Suppose that both banks choose the risky activity, such that $p_1 = p_2 = p < 1$, in which case expression (7) becomes $qpR + \beta y(1-p) = k + \tau$. In this case, the expected total repayment, which now includes the liquidation value βy in case the loan is recalled, is again equal to the principal k plus τ . Finally, note that if one bank, say bank i = 1, chooses to take on additional risk while bank i = 2 stays prudent, such that $p_1 = p < p_2 = 1$, we have from (7) that $R_1 < R_2$. The more prudent bank will thus charge a higher interest rate. Still, it is straightforward to show that the more prudent bank, namely i = 2 in this case, will attract the borrower with a higher probability ($\hat{x} < 0.5$). More formally, substituting the equilibrium loan rates R_i from (7) into (3), we obtain that¹⁹

$$\widehat{x} = \frac{1}{2} + \frac{1}{2\tau} (p_1 - p_2) \frac{1}{3} (q - \beta) y.$$
(8)

From Proposition 1 we next have the following result.

Proposition 2 For a given risk profile, as captured by the respective values of p_i , banks' expected equilibrium profits from making loans are given by

$$\pi_i = \frac{1}{2\tau} \left[\tau + \frac{y(q-\beta)(p_i - p_j)}{3} \right]^2.$$
(9)

This result gives rise to the following immediate observations. Intuitively, both banks' expected profits from the loan market are higher the more they are horizontally differentiated (as captured by a larger value of τ). Moreover, a given bank's expected profits increase in its own value p_i and decrease in its competitor's value p_j . In other words: If bank *i* is perceived by a potential borrower to be of "higher quality" as it will more likely

¹⁹Note that this also confirms that (6) indeed ensures that in equilibrium both banks are active with positive probability.

roll over the initial loan, then this allows the respective bank to realize higher profits; but if the bank's competitor is perceived to be of "higher quality", then this reduces the profits of the former bank. It is useful to collect these results more formally.

Corollary 1 We have that $d\pi_i/dp_i > 0$ and $d\pi_i/dp_j < 0$. Moreover, the likelihood with which bank i will make a loan in equilibrium is strictly increasing in p_i and strictly decreasing in p_j .

3.2 Risk-Taking Incentives

The expression for profits in Proposition 2 is now instrumental to solve for the equilibrium strategies at the beginning of period t = 1, where banks can choose whether to undertake the risky activity. We first derive a set of auxiliary comparative results, which all are immediate from undertaking the respective differentiation of expression (9).

Lemma 1 Banks' expected profits from making loans, π_i , satisfy

$$\frac{d^2\pi_i}{dp_i^2} > 0 \text{ and } \frac{d^2\pi_i}{dp_i dp_j} < 0 \text{ for } i \neq j.$$

It is worthwhile to discuss in more detail the comparative results of Lemma 1. We already know that the higher p_i , the higher are the profits of bank *i*. From $\frac{d^2\pi_i}{dp_i^2} > 0$ this effect is stronger if p_i is already high. In words, when a bank becomes more attractive to the borrower, this has a larger (positive) effect on profits if the bank is already regarded as a relatively safe choice. From $\frac{d^2\pi_i}{dp_i dp_j} < 0$, on the other hand, the effect on own profits is smaller if also the rival bank is regarded as being a safe choice (high p_j).²⁰ These comparative results are not specific to the chosen Hotelling model but hold in most other models of oligopolistic competition with *vertical* differentiation.²¹ They are also intuitive.

To see this, take first the case of $\frac{d^2\pi_i}{dp_i^2} > 0$. Recall from Corollary 1 that in equilibrium bank *i* is more likely to make a loan the higher is p_i . For instance, for i = 1 a higher value p_1 will translate into a higher value of \hat{x} . If now \hat{x} is already high, then the bank will benefit more if it can further raise R_i following a further increase in p_1 . In other words, an increase in the bank's perceived "quality", as expressed by p_1 , is thus more profitable if the bank's "market share" \hat{x} is already high, given that this allows the bank to earn a

 $^{2^{0}}$ In the formal language of Industrial Organization, the respective choices of p_i represent strategic substitutes.

 $^{^{21}}$ For a technical discussion see Athey and Schmutzler (2001).

higher margin (from a higher R_1) on a larger volume of transactions (or, more precisely, with a higher probability in our setting).

An analogous intuition applies to $\frac{d^2\pi_i}{dp_i dp_j} < 0$. An increase in, say, p_2 pushes down the threshold \hat{x} and, thereby, the "market share" of bank i = 1. By the previous arguments, this makes an increase in p_1 less profitable. Moreover, there is now an additional effect at work. As is intuitive, banks *aggregate* profits in the loan market are lower as they compete more on equal grounds, i.e., as their respective values p_i are more similar. Further below we will study in detail how these forces are affected by the prevailing degree of competition.

4 Risk Taking in Equilibrium

At t = 1, a bank's expected profits when *not* undertaking the risky activity is given by $\Pi_i = \pi_i$, where we have to substitute $p_i = 1$ into expression (9) for π_i . When undertaking activity A, instead, expected profits equal

$$\Pi_i = \pi_i + \eta_A,\tag{10}$$

where now $p_i = p$ and where η_A represents the expected payoff from A.

Intuitively, if η_A is sufficiently high, then both banks strictly prefer to undertake activity A. On the other hand, it is equally intuitive that for low values of η_A , which can also be negative, neither bank will choose activity A. In both cases, banks stay symmetric.

In contrast, a key implication of Lemma 1 is that for intermediate values of η_A the equilibrium involves an asymmetric choice: one bank becoming riskier and the other one stays safe. This follows as from Lemma 1 risk taking becomes relatively more profitable if the rival bank stays prudent, while if the rival bank is expected to undertake the risky strategy, then the safe strategy of foregoing activity A becomes relatively more attractive.²²

Proposition 3 Expressed in terms of the net present value η_A from the risky activity (A), we have the following equilibrium outcome in t = 1: There exist two thresholds $0 < \eta'_A < \eta''_A$ satisfying

²²As is standard, we restrict consideration to pure-strategy equilibria. As the outcome in a mixedstrategy equilibrium would also be asymmetric with positive probability, our insights extend, however, also to this case. Note also that for brevity's sake we do not comment separately on the (non-generic) case where η_A takes on the value of either threshold in the proposition, in which case both symmetric and asymmetric equilibria exist.

$$\eta'_A := \frac{\tau}{2} - \frac{1}{2\tau} \left[\tau - \frac{y(q-\beta)(1-p)}{3} \right]^2 \tag{11}$$

and

$$\eta_A'' := \frac{1}{2\tau} \left[\tau + \frac{y(q-\beta)(1-p)}{3} \right]^2 - \frac{\tau}{2},\tag{12}$$

such that for $\eta_A \leq \eta'_A$ neither bank chooses the risky activity, for $\eta_A \geq \eta''_A$ both banks choose the risky activity, and for $\eta'_A \leq \eta_A \leq \eta''_A$ only one bank chooses the risky activity.

Proof. See Appendix.

Proposition 3 is interesting in itself as it shows in a relatively simple model that otherwise symmetric banks may end up choosing different risk profiles for strategic reasons. For bank supervision this makes it more difficult to draw inferences on the state of the whole banking system from observations (e.g., critical incidences or site visits) at a single bank or at only few banks.

5 Risk Taking and Competition

Using Proposition 3, we can now proceed to the key comparative analysis on the impact of competition. For this recall first that the degree of horizontal differentiation is lower and competition is thus more intense as τ decreases. From differentiating the expressions for η'_A and η''_A in (11) and (12) with respect to τ , we immediately obtain the following result.

Proposition 4 As competition increases, the threshold η'_A from Proposition 3 decreases, while the threshold η''_A increases. With more intense competition it thus becomes <u>less</u> likely that both banks either take on additional risk through activity A or stay prudent, while it becomes <u>more</u> likely that only one bank chooses A.

Proposition 4 formalizes our assertion from the Introduction that more intense competition has an asymmetric effect on risk-taking incentives. Under more intense competition banks are more likely to become asymmetric in terms of the additional risk they take on, as well as in terms of their offers and market shares when competing for borrowers (cf. Corollary 1).

Clearly, for very low and very high values of η_A a change of τ will not affect the respective symmetric equilibria. An observable shift occurs instead around the respective

thresholds η'_A and η''_A , where a reduction of τ leads to a switch from a symmetric to an asymmetric equilibrium. Depending on the attractiveness of the "gambling" strategy A, more competition could thus indeed be associated both with more and with less risk taking in the banking industry. This may help to explain some of the conflicting and ambiguous empirical findings (cf. the Introduction). In the subsequent Sections we will obtain additional results in case banks are already *ex-ante* asymmetric, thereby obtaining sharper predictions on which bank will become more or less prudent.

Note finally that an intuitive expression of Proposition 4 is in terms of the interaction of vertical and horizontal differentiation. As horizontal differentiation is reduced, without vertical differentiation banks' profits on the loan market erode even if they are both (equally) attractive for a borrower. This reduces the incentives for a bank to still "play prudent" if this is also the strategy of its rival. These observations provide further intuition for the reduction of the lower threshold η'_A . On the other hand, if one bank is expected to undertake the risky strategy, then following an increase in competition it becomes more profitable for its rival to differentiate itself by being instead more prudent. This provides an intuition for the increase in the upper threshold η'_A .

6 Leverage

So far we abstracted from the impact of leverage on risk taking. This allowed to clearly work out the new effects that are at the focus of this paper. In addition, the present analysis would also be justified if banks had sufficiently low leverage. In view of the fact that banks tend to have high leverage, however, it seems warranted to consider also the case where leverage is sufficiently high to possibly create additional risk-taking incentives.²³ We show that this is indeed the case also in our model and that in this case the previous characterization of the equilibrium survives.

In more standard models, where there is no additional activity to be undertaken, there would only be three possible cash-flow realizations for the bank: zero (if a loan is made and is not repaid), k (if no loan is made), and R (if a loan is made and successfully repaid).

 $^{^{23}}$ We should note one caveat regarding our use of the term leverage in this context. In an asymmetric equilibrium, banks' expected profits will differ, implying that also the market value of their equity is different. Consequently, banks would then have *different* leverage in terms of debt value to total firm value (equity plus debt value). For simplicity we focus, however, on differences in banks' outstanding repayment obligations.

Clearly, if k exceeds the bank's repayment obligation, D_i , then a loan would be made indiscriminately. We abstract from this case by assuming that $D_i < k$. In our setting, additional cash-flow states are possible as banks can also undertake activity A. If this is undertaken but realizes a loss, the bank's cash flow equals $\beta y + f - F$ in case also a loan was made and had to be subsequently recalled. If no loan was made, the bank realizes either k + f or k + f - F, depending on the outcome of the "gamble" from strategy A, provided this was undertaken. In equilibrium, if the commercial loan market is sufficiently important (in terms of profits) compared to strategy A, the ordering of the bank's possible cash flow realizations is then as follows

$$f < \beta y + f - F < k + f - F < k + f < R + f.$$
(13)

Recall now from the Introduction that there may be two instances when the bank does not roll over the loan or extend new credit facilities. This may simply be the outcome of insolvency, or it may be necessary to prevent—in our case otherwise unavoidable—insolvency. In what follows, we want to focus (also in the light of reducing case distinctions) on the first case. We thus assume that $D_i > \beta y + f - F$: The bank can not meet its repayment obligations even if it recalls the loan to fulfill its contractual obligations under strategy $A.^{24}$ For brevity's sake we also assume symmetry with $D_i = D.^{25}$

Proposition 5 Our characterization of equilibria continues to hold if banks' repayment obligations, $D_i = D > 0$, affect their risk-taking incentives. That is, both banks undertake the risky activity if $\eta_A \ge \eta''_A$, both banks stay prudent if $\eta_A \le \eta'_A$, and only one bank undertakes the risky activity if $\eta'_A \le \eta_A \le \eta''_A$. Furthermore, competition decreases η'_A and increases η''_A . In addition, as leverage increases, both thresholds decrease, making it overall more likely that either bank undertakes the risky activity.

Proof. See Appendix.

The proof of Proposition 5 also shows also that as leverage increases, banks become more aggressive in the market for borrowers. They thus demand lower interest rates, leading to lower values of R and thus lower profits.

 $^{^{24}}$ Recall also that in this Section we want to focus, in contrast to the previous analysis, on the case of relatively high leverage.

²⁵We can extend Proposition 5 to the case of asymmetric levels of D_i , albeit this comes with a considerable extension of all involved expressions.

There are two remaining cases that must be distinguished in the proof. The most simple case is that where D < k + f - F. Here, the bank fails either if its loan was not repaid or if it undertook both activities, i.e., that of making a loan and strategy A, while the "gamble" from strategy A was not successful. If both banks stay prudent ($\eta_A \leq \eta'_A$) equityholders realize in this case an expected payoff (*net* of k) that is just equal to $\tau/2 - D$, while if both "gamble" ($\eta_A \geq \eta'_A$) their expected payoff is $\tau/2 - D + \eta_A$.²⁶ Interestingly, as the repayment obligation ("face value of debt") D increases, say by Δ_D , the value of equityholders' claims decreases by exactly the same amount, Δ_D . While the increase in the value of debt is strictly smaller than Δ_D , given that it is risky, the resulting difference is exactly made up by the reduction in each bank's overall profits, following a reduction of the interest rate that is charged to borrowers, given that banks compete more aggressively.

The second remaining case is that where D > k + f - F. Here, the bank fails also if it did not make a loan, but the gamble failed. If both banks stay prudent, then a marginal increase in D does not affect the outcome, such that equityholders still realize (*net* of k) $\tau/2 - D$. If both banks gamble, instead, then given the high probability with which debt will not be repaid in full, the value of equity now decreases by less than Δ_D (namely, by only $p\Delta_D$).

7 Heterogeneous Banks

We extend our analysis to the case where banks have already initially different risk profiles. For the purpose of our analysis we model this in a parsimonious way by stipulating that even without undertaking the risky activity A, bank i will with probability $1 - s_i$ have an outflow of funds denoted by S in t = 2. (Again, s_i is thus a measure for the quality of a bank in terms of its initial appeal to the borrower.) All that matters for our analysis is that this outflow is not too large so that βy is still sufficient to cover S + F. For simplicity, we further stipulate that both potential shocks to the bank's capital are uncorrelated. If bank i undertakes activity A, it will thus be able to roll over the loan only with probability ps_i , while without strategy A this is the case with probability s_i .

 $^{^{26}{\}rm The}$ expressions for payoffs in the asymmetric equilibrium are somewhat more complicated and contained in the Appendix.

Equilibrium Analysis. Turning first to competition in the loan market, we work, as previously, with a slightly more general notation: $p_i \in \{p, 1\}$. A borrower located at $x \in [0, 1]$ now realizes the expected profits

$$p_1 s_1 q \left(y - R_1 \right) - x \tau$$

or

$$p_2 s_2 q (y - R_2) - (1 - x) \tau$$

when obtaining a loan from bank i = 1 or bank i = 2, respectively. The critical threshold at which the borrower is just indifferent becomes thus

$$\hat{x} = \frac{1}{2} + \frac{qp_1s_1(y - R_1) - qp_2s_2(y - R_2)}{2\tau}.$$

Profits from competition on the loan market are then given by

$$\hat{x}[p_1 s_1 q R_1 + (1 - p_1 s_1) \beta y - S(1 - s_1) - k]$$
(14)

for bank i = 1, while those for bank i = 2 equal

$$(1 - \hat{x}) [p_2 s_2 q R_2 + (1 - p_2 s_2) \beta y - S (1 - s_2) - k].$$
(15)

We confine a derivation of the resulting equilibrium profits to the proof of Proposition 6. To characterize banks' incentives to undertake the risky activity A, we have to introduce some additional notation in analogy to Proposition 3. For a given bank i, there exist two thresholds for the net present value η_A of activity A: For $\eta_A \leq \eta'_{Ai}$ bank i would never want to undertake the risky activity; for $\eta_A \geq \eta''_{Ai}$ bank i would always want to undertake the risky activity; and for $\eta'_{Ai} \leq \eta_A \leq \eta''_{Ai}$ bank i would remain prudent if its rival undertakes the risky activity and would itself undertake the risky activity if its rival stays prudent instead. Clearly, if banks are initially symmetric with $s_1 = s_2$, then the thresholds for both banks are also symmetric: $\eta'_{A1} = \eta'_{A2} = \eta'_A$ and $\eta''_{A1} = \eta''_{A2} = \eta''_A$. We are thus back to the case analyzed in Proposition 3. If banks are, however, initially asymmetric, then the respective thresholds are different. In this case, we find that the initially more risky bank, say i = 2 when $s_1 > s_2$, has higher incentives to take on additional risk. Formally, both thresholds are then lower for bank i = 2: $\eta'_{A1} > \eta'_{A2}$ and $\eta''_{A1} > \eta''_{A2}$. (For a formal statement see the subsequent Proposition 6.) The intuition for this result follows immediately from our previous observations on the properties of banks' profit functions π_i (cf. Lemma 1). Recall, in particular, that π_i is convex in the probability with which bank *i* will be able to roll over the loan in t = 2 (i.e., previously p_i and now $s_i p_i$). The higher is s_i , the higher are thus also the benefits from staying prudent by not undertaking activity A, i.e., from choosing $p_i = 1$ instead of $p_i = p < 1$.

The full characterization of all equilibria depends now on the size of the difference $s_1 - s_2 > 0$. Intuitively, if this difference is sufficiently large, then in an asymmetric equilibrium, which in analogy to Proposition 3 applies for intermediary values of η_A , only the initially more risky bank i = 2 will take on additional risk. Otherwise, i.e., if $s_1 - s_2 > 0$ remains small, then for intermediary values of η_A there will still be a multiplicity of equilibria: Either bank i = 1 or bank i = 2 undertakes activity A, while the other bank stays prudent.

Proposition 6 Suppose banks have already initially different risk profiles with bank i = 2 being more risky as $s_1 > s_2$. Then banks' incentives to take on additional risk in t = 1 are given as follows:

i) If the difference in $s_1 - s_2 > 0$ is sufficiently large, then $\eta'_{A1} > \eta''_{A2}$ holds. In this case, for low net present value η_A of activity A ($\eta_A < \eta'_{A2}$) neither bank undertakes A, for high values of η_A ($\eta_A > \eta''_{A1}$) both banks undertake A, while for intermediate values of η_A ($\eta'_{A2} < \eta'_{A1}$) only the initially more risky bank, i = 1, undertakes A.

ii) If instead $\eta'_{A1} < \eta''_{A2}$ holds as the difference $s_1 - s_2 > 0$ is relatively small, then the only difference to case i) is that there now exists an interval $\eta'_{A1} < \eta_A < \eta''_{A2}$ with multiple asymmetric equilibria: Either one of the banks may undertake activity A, while the other bank does not.

Proof. See Appendix.

Competition and Risk Taking. Again, we are mainly interested in the implications of an increase in competition, as captured by a lower value of τ . We take first Case i) of Proposition 6. We can show for the interval $\eta'_{A2} < \eta_A < \eta''_{A1}$, where only bank i = 2 undertakes the additional risky activity A, that more competition reduces the lower boundary, η'_{A2} , and increases the upper boundary, η'_{A1} . (Formally, this follows immediately

from differentiating the respective thresholds in the proof with respect to τ .) In this case, we can thus unambiguously conclude that more intense competition makes it more likely *both* that the initially more risky bank, i = 2, additionally undertakes the risky activity A and that the initially less risky bank, i = 1, stays prudent.

Recall next that in Case ii) of Proposition 6 we still obtain multiple equilibria over an intermediary interval of values η_A , though the size of this interval decreases as banks become *ex-ante* more heterogeneous. If we then still choose the equilibrium where the initially more risky bank undertakes activity A, then the comparative analysis in τ from the previous Case i) still fully applies. For the comparative analysis in the following Proposition we select this case for the following reason. Suppose instead that in Case ii) and for $\eta'_{A1} < \eta_A < \eta''_{A2}$ we would select the equilibrium where only the initially *less* risky bank, i = 1, undertakes activity A. As we change η_A , our predictions for the risk-taking incentives of both banks would then be *non-monotonic* in the following way: We would predict that bank i = 2 stays prudent for very low as well as some intermediary values of the net present value η_A , while it undertakes activity A for high as well as for some lower range of values η_A . Instead, if we select in Case ii) and for $\eta'_{A1} < \eta_A < \eta''_{A2}$ the equilibrium where the initially more risky bank also undertakes activity A, then for each bank the predictions change monotonically in η_A : The respective bank will undertake activity A if and only if η_A lies above a threshold value.

Proposition 7 Suppose banks have already initially different risk profiles with bank i = 2 being more risky as $s_1 > s_2$. Then as competition increases, the initially more risky bank i = 2 becomes more likely to undertake, in addition, the risky activity A, while its initially less risky rival i = 1 is less likely to do so.

Proposition 7 would thus suggest the following policy implication. In order to identify banks that are most likely to build up additional risk, supervisory authorities should dedicate relatively more time and effort on already more risky bank as competition intensifies. In contrast, they need to worry less about initially less risky banks, for which risk-taking incentives *decrease* under more intense competition.

8 Efficiency

Our discussion so far focused on making positive predictions. We thus did not analyze whether the chosen strategies were also optimal from a social perspective.²⁷ What complicates the analysis of efficiency is the following observation. To compare total welfare, we have to consider two different sources of inefficiency that may arise from banks' decision to undertake activity A. First, this decision involves a trade-off between the present-value from this activity (if positive) and the risk of early withdrawal of funds and thus inefficient liquidation. Second, if banks end up with asymmetric profiles, then there are also inefficiencies in the loan market, given that the borrower may no longer obtain its loan at the "closest" bank, which would be the case only if $\hat{x} = 1/2$. In what follows, we take these two effects into account in two steps. Furthermore, it proves to be helpful to discuss the risk-taking incentives of the two banks in sequence, supposing first that only one bank considers to "gamble" and taking subsequently into account the incentives of the second bank.

(Efficient) Risk-taking Incentives of "First Bank" (i = 1). Suppose one bank, i = 2, is expected to stay prudent. If the other bank chooses strategy A, instead, then this generates the net present value of η_A from the risky activity, while given early withdrawal the efficiency from a newly made loan is decreased by $(1 - p)y(q - \beta)$. The crux is now, however, that by undertaking the risky activity, the bank's likelihood of being successful in the loan market will also be reduced. Precisely, starting from $\hat{x} = 1/2$ in the symmetric case, we have from (8), where $p_1 = p$ and $p_2 = 1$, that subsequently the likelihood of making a loan (or "market share") is reduced to

$$\widehat{x} = \frac{1}{2} - \frac{1}{2\tau} (1-p) \frac{1}{3} (q-\beta) y.$$
(16)

We take this into account and focus first, as noted above, solely on the benefits and costs generated by the risky activity of bank i = 1. Then, that bank i = 1 takes on the additional risk is also efficient if

$$\eta_A \ge \eta'_{Eff} := (1-p)y(q-\beta)\widehat{x},\tag{17}$$

²⁷In this paper as well as in much of the literature that analyzes banks' risk-taking incentives, we abstract from other reasons why regulators may want to ensure a more prudent behavior of banks, e.g., the risk of contagion if one bank fails.

where \hat{x} is given by (16). After some transformations we can show that $\eta'_{Eff} > \eta'_A$ holds if and only if τ is sufficiently large. In contrast, if competition is sufficiently intense, such that τ is relatively small, the opposite holds: $\eta'_{Eff} < \eta'_A$. (Cf. the proof of Proposition 8). In words, if competition is sufficiently *relaxed*, then the bank will choose "too early" to gamble. Precisely, given $\eta'_A < \eta'_{Eff}$ this holds if $\eta'_A < \eta_A < \eta'_{Eff}$, in which case the expected inefficiency from possibly having to recall the loan still outweighs the created value η_A . However, if competition is intense, then we can show that $\eta'_A > \eta'_{Eff}$. Hence, for values $\eta'_{Eff} < \eta_A < \eta'_A$ bank i = 1 should undertake the risky activity when we compare η_A with $(1 - p)y(q - \beta)\hat{x}$, but it finds it privately profitable to stay prudent, instead.

The intuition behind this somewhat surprising result is the following. With intense competition the more risky (and thus in the eyes of the potential borrower inferior) bank will be at a substantial disadvantage, in terms of the equilibrium likelihood of making a loan. For low τ this makes bank i = 1 too conservative in terms of its incentives to undertake A.

The preceding observations ignore, however, the additional welfare implications that arise from the fact that with $\hat{x} < 1/2$ there is a further loss in efficiency, given that a borrower with $x \in (\hat{x}, 1/2)$ will incur higher "shoe leather" costs than in a symmetric equilibrium. With less intense competition this reinforces the previous findings, given that we already observed that bank i = 1 had too high incentives to take on activity A. Take thus the case of more intense competition and $\eta'_{Eff} < \eta'_A$. As we show in the proof of Proposition 8, once we take into account all welfare implications, then regardless of the intensity of competition we have for the newly defined threshold $\tilde{\eta}'_{Eff}$ that $\eta'_A < \tilde{\eta}'_{Eff}$. That is, holding the "prudent" choice of bank i = 2 fixed, bank i = 1 has always too high incentives to choose the risky strategy, once total welfare is taken into account.

(Efficient) Risk-taking Incentives of "Second Bank" (i = 2). Suppose now that bank i = 1 is expected to undertake A. If also bank i = 2 undertakes the risky activity, from a welfare perspective we again have to first trade-off again the realization of η_A with the fact that the loan may be inefficiently recalled. Taking the threshold \hat{x} from (16), the latter inefficiency is equal to $1 - \hat{x}$ times $(1-p)y(q-\beta)$. From this we obtain the criterion

$$\eta_A \ge \eta_{Eff}'' := (1-p)y(q-\beta)(1-\hat{x}),$$

where η''_{Eff} is clearly strictly higher than the threshold η'_{Eff} in (17). Moreover, we now find that $\eta''_A < \eta''_{Eff}$ always holds. That is, when only trading off η_A with the possibility of an inefficient termination of a loan, then the transition to a symmetric equilibrium where both banks undertake the risky activity is always too "early". However, when we consider total welfare, i.e., again including the "shoe leather" costs of the borrower, then as competition becomes sufficiently relaxed, results are reversed: Bank i = 2 has insufficient incentives to take on additional risk.

Summary. The following Proposition summarizes results and provides a comparison between the efficiency benchmark and the equilibrium outcome. For brevity we restrict the statement of the Proposition to the full-welfare benchmark.

Proposition 8 Take as a benchmark that of total welfare, including the NPV of strategy A, inefficiencies from early liquidation of a funded project, and possibly inefficiently high ("shoe leather") costs in the market for borrowers. Then, there are two thresholds $\tilde{\eta}'_{Eff} < \tilde{\eta}''_{Eff}$ such that no bank should undertake the risky activity for $\eta_A \leq \tilde{\eta}'_{Eff}$, both banks for $\eta_A \geq \tilde{\eta}''_{Eff}$, and only one bank for $\tilde{\eta}'_{Eff} \leq \eta_A \leq \tilde{\eta}''_{Eff}$. These thresholds compare with the equilibrium outcome in Proposition 3 as follows:

i) It always holds that $\eta'_A < \tilde{\eta}'_{Eff}$: The "first" bank that chooses the risky strategy will always do so "too early", i.e., for values of η_A that are still too low from a welfare perspective.

ii) In contrast, $\eta''_A < \tilde{\eta}''_{Eff}$ holds only if competition is not too intense (low τ), while otherwise (high τ) it holds that $\eta''_A > \tilde{\eta}''_{Eff}$. It thus depends on competition whether for interim values of η_A there is too much risk taking as both banks choose A, but should not, or too little risk taking as both banks should choose A, but only one bank does so.

Proof. See Appendix.

A key observation from Proposition 8, as well as from our preceding discussion, is that even in the absence of (additional) risk-taking incentives from leverage, banks may not have first-best incentives to take on risk. However, as our discussion and Proposition 8 reveal, risk-taking incentives can be too high *or* too low, depending on how a bank's concern for its reputation with potential corporate borrowers interacts with competition. Here, in line with our preceding observations on risk-taking incentives, from Proposition 8 we obtain ambiguous results if competition becomes sufficiently intense. In this case, there may be either too much (namely, for low values of η_A) or too little (namely, for high values of η_A) risk taking in the market.²⁸

Admittedly, Proposition 8 does not generate clear-cut prescriptions for regulation and supervision. Also, recall that our welfare analysis neglects any externalities that arise from risk-taking and the resulting possible failure of a bank. Still, however, Proposition 8 and the preceding analysis point to the key *beneficial* role that competition can play to mitigate risk-taking incentives. In particular, when one bank is expected to engage in additional risk taking, with intense competition a rival bank has much to gain when it stays "prudent" and, thereby, enhances its attractiveness in the eyes of commercial borrowers. This mechanism is, however, only at work when competition prevails.²⁹

9 Conclusion

This paper analyzes how a bank's reputation with potential corporate borrowers interacts with its incentives to take on (additional) risk, e.g., through proprietary trading, additional loan commitments, guarantees, or other contingent obligations. Corporate borrowers, in particular those who find it difficult to borrow from the market or from other lenders at short notice, care about the bank's overall financial strength, as this will determine their future access to credit. Banks that face a short-fall of profits or that are hit by high payment obligations will cease to roll over outstanding loans or they will call back existing credit facilities.

We explore the model to investigate primarily the interaction between risk-taking incentives and competition. Though much ink has already been spilled on this topic (cf. the Introduction for an account of the literature), a key distinctive feature of our analysis is that it generates an *asymmetric* response of banks' risk-taking incentives to competition. In particular, we found that banks that are already more riskier have an increasing appetite to take on more risk as competition intensifies, while their safer rivals are more likely to stay prudent. As noted in the Introduction, this may shed further light on some conflicting and ambiguous findings in the literature.

²⁸That applies, more precisely, for values of η_A that are low but not too low, and values of η_A that are high but not too high–as, otherwise, no bank or both banks would and should take on additional risk.

²⁹In the present (as of spring 2009) financial crisis it becomes, indeed, clear that banks differ much in the (additional) risks that they have engaged in, which now translates into key competitive differences.

In addition, our findings provide some guidance for bank supervisors and regulators. Supervisors' response to increased competition should be asymmetric, by targeting already riskier banks, while relaxing oversight of presently more prudent institutions (provided that the objective is to identify and potentially limit (additional) risk taking). Furthermore, regulators and supervisors should view competition not as being detrimental to financial stability. Instead, when banks' risk positions are transparent to commercial borrowers, we argue that competition may stifle banks' risk appetite, as it turns "prudence" into a key competitive variable.

10 Appendix

Proof of Proposition 3. Using (10), if both banks undertake activity A, then each bank realizes the profit $\eta_A + \frac{\tau}{2}$. A deviating bank obtains, instead, the profit

$$\frac{1}{2\tau} \left[\tau + \frac{(qy - \beta y)(1 - p)}{3}\right]^2.$$
(18)

Comparing the two profit levels shows that we can support a symmetric equilibrium with risk taking if and only if $\eta_A \ge \eta''_A$, as characterized in (12). If both banks are prudent, they realize profits of $\frac{\tau}{2}$. If one bank deviates and undertakes activity A instead, its profit equals

$$\eta_A + \frac{1}{2\tau} \left[\tau - \frac{(qy - \beta y)(1 - p)}{3} \right]^2.$$
(19)

Comparing the two profit levels shows now that we can support a symmetric equilibrium without risk taking if and only if $\eta_A \leq \eta'_A$, as characterized in (11).

Turning finally to asymmetric pure-strategy equilibria, note first that the respective profits of the prudent and the gambling bank are then given by (18) and (19), respectively. Comparing these profits to the profits $\frac{\tau}{2}$ and $\eta_A + \frac{\tau}{2}$ under a symmetric choice, i.e., if one bank deviates, we obtain that the bank undertaking A will not deviate if $\eta_A \leq \eta'_A$, while the prudent bank will stay so only if $\eta_A \geq \eta''_A$. Observe finally that $\eta''_A > \eta'_A$.

Proof of Proposition 5. Note first that there are two cases to be distinguished: that with $\beta y - F + f < D < k + f - F$ and that with k + f - F < D < k.

The Case with $\beta y - F + f < D < k + f - F$. Using (3) for \hat{x} , the expected payoff to equityholders of bank i = 1 is given by

$$\hat{x}\left(p_{1}q\left(R_{1}+f_{1}-D_{1}\right)\right)+\left(1-\hat{x}\right)\left(p_{1}\left(k+f_{1}-D_{1}\right)+\left(1-p_{1}\right)\left(k+f_{1}-F-D_{1}\right)\right),\quad(20)$$

where $p_1 = p$ and $f_1 = f$ if the bank chooses activity A, while otherwise $p_1 = 1$ and $f_1 = 0$. Bank i = 1 chooses R_1 to maximize (20). For bank i = 2 we obtain likewise for equityholders' payoff

$$(1 - \hat{x}) \left(p_2 q \left(R_2 + f_2 - D_2 \right) \right) + \hat{x} \left(p_2 \left(k + f_2 - D_2 \right) + (1 - p_2) \left(k + f_2 - F - D_2 \right) \right).$$
(21)

From the respective first-order conditions for (20) and (21) we obtain after some transformations (and in case of an interior solution for \hat{x}) that

$$R_{i}p_{i}q = \tau + k - (D - f_{i})(1 - qp_{i}) - F(1 - p_{i}) + \frac{(f_{j} - f_{i})(1 - qp_{i}) + (p_{i} - p_{j})(q(y - D) + qf_{j} - F)}{3},$$

which after substitution into (20) and (21), respectively, yields the expected payoff to equityholders of bank i

$$f_{i} - F(1 - p_{i}) - D$$

$$+ \frac{1}{2\tau} \left(\tau + \frac{(f_{j} - f_{i})(1 - qp_{j}) + (p_{i} - p_{j})(q(y - D) - F + qf_{i})}{3} \right)^{2}.$$
(22)

If both banks undertake activity A, then each bank's equityholders' payoff is calculated by inserting $p_i = p_j = p$ and $f_i = f_j = f$ into (22), yielding

$$\frac{\tau}{2} - D + \eta_A. \tag{23}$$

A deviating bank's equityholders' payoff is, instead, after substitution of $p_i = 1$, $p_j = p$, $f_i = 0$, and $f_j = f$, equal to

$$\frac{1}{2\tau} \left(\tau + \frac{pf(1-q) + (1-p)(q(y-D) + f - F)}{3} \right)^2 - D.$$
 (24)

Comparing this to (23) shows that we can support a symmetric equilibrium with risk taking if and only if

$$\eta_A \ge \widehat{\eta}_A'' := \frac{1}{2\tau} \left(\tau + \frac{pf(1-q) + (1-p)(q(y-D) + f - F)}{3} \right)^2 - \frac{\tau}{2}.$$

If both banks are prudent $(p_i = p_j = 1 \text{ and } f_i = f_j = 0)$, their equityholders' payoff is $\frac{\tau}{2} - D$. If one bank deviates and undertakes activity A instead $(p_i = p, p_j = 1 \text{ and} f_i = f, f_j = 0)$, its equityholders' payoff equals

$$\eta_A - D + \frac{1}{2\tau} \left(\tau - \frac{f(1-q) + (1-p)(q(y-D) + fq - F)}{3} \right)^2.$$
 (25)

Comparing the two payoffs shows now that we can support a symmetric equilibrium without risk taking if and only if

$$\eta_A \le \hat{\eta}'_A := \frac{\tau}{2} - \frac{1}{2\tau} \left(\tau - \frac{\left(pf\left(1-q\right) + \left(1-p\right)\left(q\left(y-D\right) + f - F\right)\right)}{3} \right)^2.$$

Turning to asymmetric pure-strategy equilibria, note first that the payoffs of the equityholders of the prudent and the risk taking bank are then given by (24) and (25), respectively. Comparing these to $\frac{\tau}{2} - D$ and $\frac{\tau}{2} - D + \eta_A$ under a symmetric choice, we obtain an asymmetric equilibrium if $\hat{\eta}'_A \leq \eta_A \leq \hat{\eta}''_A$, where indeed $\hat{\eta}''_A > \hat{\eta}'_A$.

The Case with k + f - F < D < k. If k + f - F < D < k, using (3) the expected payoff of the equityholders' of bank i = 1 is given by

$$\hat{x} \left(p_1 q \left(R_1 + f_1 - D \right) \right) + \left(1 - \hat{x} \right) \left(p_1 \left(k + f_1 - D \right) \right).$$
(26)

Likewise, the optimal choice of R_2 for the rival bank i = 2 maximizes

$$(1 - \hat{x}) \left(p_2 q \left(R_2 + f_2 - D \right) \right) + \hat{x} \left(p_2 \left(k + f_2 - D \right) \right).$$
(27)

From the respective first-order conditions for (26) and (27) we have that

$$p_{i}qR_{i} = \tau + kp_{i} - p_{i}(1-q)(D-f_{i}) + \frac{(p_{i} - p_{j})(qy-k) + (1-q)(p_{i}(D-f_{i}) - p_{j}(D-f_{j}))}{3}.$$
(28)

Using expressions (26) and (28), we can calculate equityholders' equilibrium payoffs:

$$\frac{1}{2\tau} \left(\tau + \frac{(p_i - p_j)(qy - k) + (1 - q)(p_i(D - f_i) - p_j(D - f_j))}{3}\right)^2 - (1 - p_i)k - p_i(D - f_i).$$
(29)

If both banks undertake activity A, then each bank's equityholders' payoff is calculated by inserting $p_i = p_j = p$ and $f_i = f_j = f$ into (29), which yields

$$\frac{\tau}{2} + \eta_A + (F - f)(1 - p) - k(1 - p) - pD.$$
(30)

In case of a deviation, such that $p_i = 1$, $p_j = p$, $f_i = 0$, and $f_j = f$, we have that

$$\frac{1}{2\tau} \left(\tau + \frac{(1-p)(qy-k) + (D(1-p) + fp)(1-q)}{3} \right)^2 - D.$$
(31)

Comparing the two payoffs shows that we can support a symmetric equilibrium with risk taking if and only if

$$\eta_A \geq \widetilde{\eta}_A'' := (1-p)\left(-F + f + k - D\right) - \frac{\tau}{2} + \frac{1}{2\tau} \left(\tau + \frac{pf\left(1-q\right) + (1-p)\left(D\left(1-q\right) + qy - k\right)}{3}\right)^2.$$

If both banks are prudent $(p_i = p_j = 1 \text{ and } f_i = f_j = 0)$, their equityholders' payoff is $\frac{\tau}{2} - D$. If one bank deviates and undertakes activity A instead, such that $p_i = p$, $p_j = 1$, $f_i = f'$, and $f_j = 0$, its equityholders' payoff equals

$$\frac{1}{2\tau} \left(\tau - \frac{(1-p)(qy-k) + (D(1-p) + fp)(1-q)}{3} \right)^2 - k(1-p) - pD.$$
(32)

Comparing the two payoffs shows now that we can support a symmetric equilibrium without risk taking if and only if

$$\eta_A \leq \tilde{\eta}'_A := (1-p)\left(-F + f + k - D\right) + \frac{\tau}{2} \\ -\frac{1}{2\tau} \left(\tau - \frac{pf(1-q) + (1-p)\left(D\left(1-q\right) + qy - k\right)}{3}\right)^2.$$

Turning to asymmetric pure-strategy equilibria, note first that the payoffs of the equityholders' of the prudent and the gambling bank are then given by (31) and (32), respectively. Comparing these to $\frac{\tau}{2} - D$ and $\frac{\tau}{2} + \eta_A + (F - f)(1 - p) - k(1 - p) - pD$ under a symmetric choice, we can support an asymmetric outcome if $\tilde{\eta}'_A \leq \eta_A \leq \tilde{\eta}''_A$, where we use that $\tilde{\eta}''_A > \tilde{\eta}'_A$.

Proof of Proposition 6. From the respective first-order conditions we obtain the following results.³⁰ If banks have already initially a different risk profile, as expressed by the probabilities $1 - s_i$ with which they will have to recall a loan in t = 2 even without strategy A, then their expected profits from the loan market are given by

$$\pi_{i} = \frac{1}{2\tau} \left(\tau + \frac{y \left(p_{i} s_{i} - p_{j} s_{j} \right) \left(q - \beta \right)}{3} \right)^{2} - S \left(1 - s_{i} \right).$$
(33)

Note next that we stipulate without loss of generality that $s_1 > s_2$. Suppose now that one bank is anticipated to undertake activity A. If the other bank, say now i = 1, also undertakes activity A, then from (33) the respective profits equal

$$\eta_A + \frac{1}{2\tau} \left(\tau + \frac{yp(s_1 - s_2)(q - \beta)}{3} \right)^2 - S(1 - s_1),$$

while if bank i = 1 stays prudent profits are

$$\frac{1}{2\tau} \left(\tau + \frac{y \left(s_1 - p s_2 \right) \left(q - \beta \right)}{3} \right)^2 - S \left(1 - s_1 \right).$$

Comparing these two profits, we have that if bank i = 1 expects its rival to undertake A, then it prefers to do so as well if

$$\eta_A \ge \eta_{A1}'' := \frac{1}{6\tau} y \left(q - \beta \right) s_1 (1-p) \left(\frac{y \left(q - \beta \right)}{3} \left(s_1 (1+p) - 2ps_2 \right) + 2\tau \right).$$

Proceeding likewise for bank i = 2, we obtain the threshold

$$\eta_A \ge \eta_{A2}'' := \frac{1}{6\tau} y \left(q - \beta \right) s_2 (1 - p) \left(\frac{y \left(q - \beta \right)}{3} \left(s_2 (1 + p) - 2ps_1 \right) + 2\tau \right).$$

Note that when subtracting $\eta_{A2}^{\prime\prime}$ from $\eta_{A1}^{\prime\prime}$, we obtain

$$\frac{1}{3}y(q-\beta)(s_1-s_2)(1-p) > 0.$$

³⁰Again, to obtain in equilibrium an interior solution $0 < \hat{x} < 1$ it must hold that $|y(\beta - q)(p_2s_2 - p_1s_1)| < 3\tau$, which in turn is always satisfied as long as $y(q - \beta)(s_1 - ps_2) < 3\tau$.

Suppose next that a bank's rival is anticipated not to undertake activity A. If the other bank, say again first i = 1, now undertakes activity A alone, then from (33) the respective profits equal

$$\eta_A + \frac{1}{2\tau} \left(\tau + \frac{y \left(p s_1 - s_2 \right) \left(q - \beta \right)}{3} \right)^2 - S \left(1 - s_1 \right),$$

while if bank i = 1 also stays prudent profits are

$$\frac{1}{2\tau} \left(\tau + \frac{y(s_1 - s_2)(q - \beta)}{3} \right)^2 - S(1 - s_1).$$

Comparing these two profits, we have that bank i = 1 prefers to also undertake A if

$$\eta_A \ge \eta'_{A1} := \frac{1}{6\tau} y \left(q - \beta \right) s_1 (1 - p) \left(\frac{y \left(q - \beta \right)}{3} \left(s_1 (1 + p) - 2s_2 \right) + 2\tau \right),$$

which satisfies $\eta'_{A1} < \eta''_{A1}$. Proceeding likewise for bank i = 2, we obtain the threshold

$$\eta_A \ge \eta'_{A2} := \frac{1}{6\tau} y \left(q - \beta \right) s_2 (1-p) \left(\frac{y \left(q - \beta \right)}{3} \left(s_2 (1+p) - 2s_1 \right) + 2\tau \right).$$

Note again that when subtracting η'_{A2} from η'_{A1} , we obtain

$$\frac{1}{3}y(q-\beta)(s_1-s_2)(1-p) > 0.$$

Note finally that $\eta'_{A1} > \eta''_{A2}$ holds in case

$$s_1 - s_2 > \frac{3s_1s_2(1-p)}{y(q-\beta)\tau}.$$

The characterization of the different equilibria for cases i) and ii) follows then immediately from the construction of the different thresholds. \blacksquare

Proof of Proposition 8. It is convenient to define for this proof $B := (1 - p)y(q - \beta)$. The threshold (17) then becomes, after substitution from (16),

$$\eta_{Eff}' = \frac{1}{2}B\left(1 - \frac{1}{3\tau}B\right),\,$$

while we can use from (11) that

$$\eta'_A := \frac{1}{2} \left[\tau - \frac{1}{\tau} \left(\tau - \frac{B}{3} \right)^2 \right].$$

That $\eta'_{Eff} > \eta'_A$ follows then if

$$B\left(1-\frac{1}{3\tau}B\right) > \tau - \frac{1}{\tau}\left(\tau - \frac{B}{3}\right)^2,$$

which finally transforms to $B < 3\tau/2$. (Note that this is *not* implied by (6), which only requires that $B < 3\tau$.)

We next take also into account the "shoe leather" costs of the borrower:

$$E(x\tau) = C := \tau \left(\int_0^{\widehat{x}} x dx + \int_0^{1-\widehat{x}} x dx \right) = \tau \left[\frac{1}{2} - \widehat{x}(1-\widehat{x}) \right]$$

We have $C = \tau/4$ in a symmetric equilibrium. Given $\hat{x} = \frac{1}{2} - \frac{1}{6\tau}B$ for an asymmetric equilibrium, the respective costs become

$$C = \tau \frac{1}{4} \left(1 + \frac{B^2}{9\tau^2} \right). \tag{34}$$

The adjusted efficiency threshold then becomes

$$\widetilde{\eta}'_{Eff} := \frac{1}{2}B\left(1 - \frac{1}{3\tau}B\right) + \tau \frac{1}{4}\left(1 + \frac{B^2}{9\tau^2}\right),$$

such that $\tilde{\eta}'_{Eff} > \eta'_A$ holds only if

$$\tau > \left(\frac{B}{3\tau}\right) \left(B - 2\tau\right). \tag{35}$$

Note now that we have from (6) that $\tau > B/3$, implying that (35) always holds if $B - 2\tau < \tau$, which once again becomes $\tau > B/3$.

To compare $\eta''_{Eff} = B(1 - \hat{x})$ with η''_A , note that $\eta''_{Eff} > \eta''_A$ holds if

$$B\left(1+\frac{1}{3\tau}B\right) > \frac{1}{\tau}\left(\tau+\frac{B}{3}\right)^2 - \tau,$$

which after some transformations indeed always holds. Using next C from (34) for the asymmetric case, we obtain a new threshold

$$\widetilde{\eta}_{Eff}^{\prime\prime} := \frac{1}{2} B \left(1 + \frac{1}{3\tau} B \right) - \tau \frac{1}{4} \left(1 + \frac{B^2}{9\tau^2} \right),$$

for which $\widetilde{\eta}_{Eff}'' > \eta_A''$ holds if

$$\tau < \frac{B}{3\tau} \left(2\tau + B \right). \tag{36}$$

Note that this is compatible with $\tau > B/3$ from (6). Substituting $\tau = B/3$, the condition is satisfied up to some threshold for τ , from which on the converse of (6) holds strictly.

References

- Acharya, V. and Yorulmazer, T., Too Many to Fail An Analysis of Timeinconsistency in Bank Closure Policies, Journal of Financial Intermediation 16 (2007), 1-31.
- [2] Athey, S. and Schmutzler, A., Investment and Market Dominance, The RAND Journal of Economics 32, 1 (2001), 1-26.
- [3] Beck, T., Demirgüc-Kunt, A., and Levine, R., Bank Concentration and Crises, NBER Working Paper No. 9921 (2003).
- [4] Besanko, D. and Thakor, A., Relationship Banking, Deposit Insurance and Bank Portfolio Choice, in C. Mayer and X. Vives, eds., Capital Markets and Financial Intermediation, Cambridge Univ. Press (1993).
- [5] Boot, A. W. A., Relationship Banking: What Do We Know?, Journal of Financial Intermediation 9 (2000), 7-25.
- [6] Boot, A. W. A. and Marinč, M., Competition and Entry in Banking: Implications for Stability and Capital Regulation, ACLE Working Paper No. 2005-07 (2006).
- [7] Boot, A. W. A. and Thakor, A. V., Can Relationship Banking Survive Competition?, The Journal of Finance 55 (2000), 679-713.
- [8] Boyd, J. H. and De Nicolo, G., The Theory of Bank Risk-Taking and Competition Revisited, The Journal of Finance 60 (2005), 1229-1343.
- [9] Boyd, J. H., De Nicolo, G., and Jalal, A. M., Bank Risk-Taking and Competition Revisited: New Theory and New Evidence, IMF Working Paper No. 06/297 (2006).
- [10] Caminal, R. and Matutes, C., Can Competition in the Credit Market be Excessive?, CEPR Discussion Paper No. 1725 (1997).
- [11] Carletti, E. and Hartmann, P., Competition and Financial Stability. What's Special about Banking? in P. Mizen, ed., Monetary History, Exchange Rates and Financial Markets: Essays in Honor of Charles Goodhart, Vol. 2, Edward Elgar, Cheltenham/UK (2003).
- [12] Chan, Y., Greenbaum, S., and Thakor, A., Information Reusability, Competition and Bank Asset Quality, Journal of Banking and Finance 10 (1986), 243-253.
- [13] Cordella, T. and Yeyati, E. L., Financial Opening, Deposit Insurance, and Risk in a Model of Banking Competition, European Economic Review 46 (2002), 471–485.
- [14] Degryse, H. and Ongena, S., Distance, Lending Relationships, and Competition, Journal of Finance 60 (2005), 231-266.
- [15] Degryse, H., Laeven, L., and Ongena, S., The Impact of Organizational Structure and Lending Technology on Banking Competition, mimeo (2006).

- [16] De Nicoló, G. and Loukoianova, E., Bank Ownership, Market Structure and Risk, mimeo (2007).
- [17] Detragiache E., Garella P., and Guiso L., Multiple vs. Single Banking Relationships: Theory and Evidence, The Journal of Finance 55 (2000), 1133-1161.
- [18] Djankov, S., Jindra, J., and Klapper, L. F., Corporate Valuation and the Resolution of Bank Insolvency in East Asia, Journal of Banking & Finance 29, 8-9 (2005), 2095-2118.
- [19] Gropp, R., Hakenes, H., and Schnabel, I., Competition, Risk-Shifting, and Public Bail-Out Policies, mimeo (2006).
- [20] Hellmann, T. F., Murdock, K. C., and Stiglitz, J. E., Liberalization, Moral Hazard in Banking, and Prudential Regulation: Are Capital Requirements Enough?, The American Economic Review 90, 1 (2000), 147-165.
- [21] Jiménez, G., Lopez, J. A., and Saurina, J., How Does Competition Impact Bank Risk-Taking?, mimeo (2007).
- [22] Keeley, M. C., Deposit Insurance, Risk and Market Power in Banking, The American Economic Review 80 (1990), 1183-1200.
- [23] Martinez-Miera, D. and Repullo, R., Does Competition Reduce the Risk of Bank Failure?, CEMFI Working Paper No. 0801 (2008).
- [24] Matutes, C. and Vives, X., Competition for Deposits, Fragility, and Insurance, Journal of Financial Intermediation 5 (1996), 184-216.
- [25] Matutes, C. and Vives, X., Imperfect Competition, Risk Taking, and Regulation in Banking, European Economic Review 44 (2000), 1-34.
- [26] Perotti, E. C. and Suarez, J., Last Bank Standing: What Do I Gain If You Fail?, European Economic Review 46 (2002), 1599-1622.
- [27] Repullo, R., Capital Requirements, Market Power, and Risk-Taking in Banking, Journal of Financial Intermediation 13 (2004), 156-182.
- [28] Slovin, M. B., Sushka, M. E., and Polonchek, J. A., The Value of Bank Durability: Borrowers as Bank Stakeholders, The Journal of Finance 48 (1993), 247-266.
- [29] Suarez, J., Closure Rules, Market Power and Risk-Taking in a Dynamic Model of Bank Behavior, Discussion Paper 196, LSE/FMG (1994).
- [30] Stein, J. C., An Adverse-Selection Model of Bank Asset and Liability Management with Implications for the Transmission of Monetary Policy, The RAND Journal of Economics 29, 3 (1998), 466-486.
- [31] Winton, A., Institutional Liquidity Needs and the Structure of Monitored Finance, The Review of Financial Studies 16, 4 (Winter, 2003), 1273-1313.

[32] Yamori, N. and Murakami, A., Does Bank Relationship Have an Economic Value? -The Effect of Main Bank Failure on Client Firms, Economics Letters 65, 1 (1999), 115-120.

WORKING PAPERS

1 (2006)	Helmut Siekmann	The Burden of an Ageing Society as a Public Debt (veröffentlicht in: European Public Law 2007 (13/3))
2 (2006)	Helmut Siekmann	Die Unabhängigkeit von EZB und Bundesbank nach geltendem Recht und dem Vertrag über eine Verfassung für Europa
3 (2006)	Helmut Siekmann	Die Verwendung des Gewinns der Europäischen Zentralbank und der Bundesbank
4 (2006)	Reinhard H. Schmidt Aneta Hryckiewicz	Financial Systems - Importance, Differences and Convergence
5 (2006)	Roman Inderst Holger M. Mueller Felix Münnich	Financing A Portfolio of Projects
6 (2006)	Roman Inderst Holger M. Mueller	A Lender-Based Theory of Collateral
7 (2006)	Joachim Wieland	Staatsverschuldung als Herausforderung für die Finanzverfassung (veröffentlicht in: JZ 2006, S. 751 ff.)
8 (2007)	Helmut Siekmann	Der Anspruch auf Herstellung von Transparenz im Hinblick auf die Kosten und Folgekosten der Steinkohlesubventionierung und den Börsengang der RAG AG
9 (2007)	Henry Ordower	Demystifying Hedge Funds: A Design Primer (veröffentlicht in: UC Davis Business Law Journal 2007 (7/2), S. 323-372)
10 (2007)	Helmut Siekmann	Die Spielbankabgabe und die Beteiligung der Gemeinden an ihrem Aufkommen – zugleich ein Beitrag zu den finanzverfassungsrechtlichen Ansprüchen der Gemeinden (veröffentlicht in: Organisation und Verfahren im sozialen Rechtsstaat, Festschrift für Friedrich E. Schnapp zum

70. Geburtstag, Herausgegeben von Hermann Butzer,
Markus Kaltenborn, Wolfgang Meyer, 2008, S.319-345)

11 (2007)	Symposium am 26.11.2007 in Frankfurt am Main	Neuordnung der föderalen Finanzbeziehungen
12 (2007)	Stefan Gerlach Peter Kugler	Deflation and Relative Prices: Evidence from Japan and Hong Kong
13 (2007)	Katrin Assenmacher- Wesche Stefan Gerlach Toshitaka Sekine	Monetary Factors and Inflation in Japan
14 (2007)	Guntram B. Wolff	Schuldenanstieg und Haftungsausschluss im deutschen Föderalstaat: Zur Rolle des Moral Hazard
15 (2008)	Helmut Siekmann	Föderalismuskommission II für eine zukunftsfähige Gestaltung der Finanzsystem nutzen
16 (2008)	Katrin Assenmacher- Wesche Stefan Gerlach	Ensuring Financial Stability: Financial Structure and the Impact of Monetary Policy on Asset Prices
17 (2008)	Helmut Siekmann	Stellungnahme für die öffentliche Anhörung des Haushaltsausschusses zu dem Gesetzentwurf der Fraktion der SPD und Bündnis 90/Die Grünen für ein Gesetz zur Änderung der Hessischen Landeshaushaltsordnung
18 (2008)	Hans Genberg Cho-Hoi Hui	The credibility of <i>The Link</i> from the perspective of modern financial theory
19 (2009)	Helmut Siekmann	Stellungnahme für die öffentliche Anhörung des Ausschusses für Wirtschaft, Mittelstand und Energie und des Haushalts- und Finanzausschusses des Landtags Nordrhein-Westfalen Keine Hilfe für Banken ohne einen neuen Ordnungsrahmen für die Finanzmärkte
20 (2009)	Chun-Yu Ho Wai-Yip Alex Ho	On the Sustainability of Currency Boards: Evidence from Argentina and Hong Kong

21 (2009)	Stefan Gerlach	The Risk of Deflation
22 (2009)	Tim Oliver Berg	Cross-country evidence on the relation between equity prices and the current account
23 (2009)	Melanie Döge Stefan Jobst	Aktienrecht zwischen börsen- und kapitalmarktorientiertem Ansatz
24 (2009)	Helmut Siekmann	Die Schaffung von Einrichtungen der Finanzaufsicht auf EU-Ebene Stellungnahme zu dem Vorschlag der Sachverständigengruppe unter dem Vorsitz von Jacques de Larosière
25 (2009)	Helmut Siekmann	Die Neuordnung der Finanzmarktaufsicht
26 (2009)	Helmut Siekmann	Stabilisierung der WestLB AG durch Garantien des Landes NRW Stellungnahme für die öffentliche Anhörung des Haushalts- und Finanzausschusses des Landtags Nordrhein-Westfalen am 29. Oktober 2009
27 (2009)	Roman Inderst	Loan Origination under Soft- and Hard-Information Lending
28 (2009)	Hasan Doluca Roman Inderst Ufuk Otag	Bank Competition and Risk-Taking When Borrowers Care about Financial Prudence